Author's response to reviews

Title: All-arthroscopic versus mini-open repair of small or moderate-sized rotator cuff tears: A protocol for a randomized trial [NCT00128076] - A study protocol

Authors:

Joy C MacDermid (macderj@mcmaster.ca)
Richard Holtby (richard.holtby@sw.ca)
Helen Razmjou (helen.razmjou@sw.ca)
Dianne Bryant (jackowdm@mcmaster.ca)
Joints Canada (jointscanada@jointscanada.ca)

Version: 5 Date: 3 March 2006

Author's response to reviews: see over
MS: 1330702885779378
All-Arthroscopic versus Mini-Open Repair of Small or Moderate Sized Rotator Cuff Tears: A Protocol for a Randomized Trial [NCT00128076] - A study protocol
Joy C MacDermid, Richard Holtby, Helen Razmjou, Dianne Bryant and Joints Canada
BMC Musculoskeletal Disorders
Subject: Formatting Changes for MS: [1330702885779378]

Formatting changes requested
---------------------------------
Website addresses - Please cite website addresses in the reference list and remove them from the manuscript text. Cite them using the following format:

Website title [http://www.website.com].
Where possible, please avoid using very long addresses. Please remove the website addresses from the title page.

**JOINTS website address removed from title page and e-mail address (jointscanada@jointscanada.ca) added to title page.**

Abstract - Abstracts must be structured into Background, Methods/Design, Discussion. Please remember to also update the Abstract details on the submission page. Please add a discussion section in the manuscript and alter the the submission page to include Background Methods/Design and Discussion.

**Discussion has been added to the abstract. Abstract in the submission page has been updated to include discussion.**